ACM Email Room

Participant PC Setup

1. Write down all your current settings in the spaces below. You will need this information when you go home.

2. On the start menu, proceed to Settings and select Control Panel.

Select the Network icon. 3. Under Protocols, select TCP/IP and click on the Properties button.

4. Using the table below, select the tabs and change the information after recording your present settings.

5. After entering the ACM settings, click OK. If you are instructed to reboot your computer, do so. Keep these settings for the duration of the conference.

6. You may now plug into the ACM net on any available cable in the email room. The room is open from 8 AM to 6:30 PM each day.

Tab Instructions Number Original ACM IP Insert IP address IP ___.___.___.___ 132.236.46.___ IP Insert subnet mask Subnet mask ___.___.___.___ 255.255.255.0 Gateway Remove installed

gateways and insert Gateway ___.___.___.___ 132.236.46.1 ACM gateway DNS Enable DNS and DNS servers ___.___.___.___ 128.84.242.46 add ACM name ___.___.___.___ 128.35.82.50 servers ___.___.___.___ ___.___.___.___